Author's response to reviews

Title: Elevated C1orf63 expression is correlated with CDK10 and predicts better outcome for advanced breast cancers: a retrospective study

Authors:

Chaoqun Hong (chaoqunhong@gmail.com)
Fan Zhang (lemon_fan@163.com)
Yanjie You (youyanjie@163.com)
Armando E Giuliano (armando.giuliano@cshs.org)
Xiaojian Cui (xiaojian.cui@cshs.org)
Guojun Zhang (guoj_zhang@yahoo.com)
Yukun Cui (ycui27@gmail.com)

Version: 4  Date: 25 December 2014

Author's response to reviews: see over
Cover Letter

Dear Editors,

We enclose a manuscript entitled “Elevated C1orf63 expression is correlated with CDK10 and predicts better outcome for advanced breast cancers: a retrospective study” for publication in *BMC Cancer*. The main text includes 24 pages, 4 tables, and 1 figure prepared according to the journal's Instructions to Authors. The supplementary material with 3 supplementary tables and 2 supplementary figures is also attached.

As far as we know, C1orf63 has been seldomly studied in breast cancer. Our paper primarily investigated the expression and clinical implication of C1orf63 in breast cancer, and the results indicated the bipolar characters of C1orf63 in the progression of breast cancer. Since elevated expression was observed in tumors, C1orf63 might contribute to breast cancer oncogenesis. However, C1orf63 predicted better prognosis for advanced TNM stage breast cancers, and was correlated with the cell cycle-dependent gene, CDK10.

We promise that the content has not been published or submitted for publication elsewhere, and all the authors listed have contributed significantly and are in agreement with the submission of the manuscript. No potential conflicts of interest were disclosed.

We respectfully submit four potential suitable peer reviewers based on their expertise in the field:

**Magdalena Chechlinska**, *Department of Immunology, Maria Sklodowska-Curie Memorial Cancer Centre and Institute of Oncology, Warsaw, Poland.*
E-mail: chech@coi.waw.pl

**Dietmar Pils**, *Department of Obstetrics and Gynecology, Molecular Oncology Group, Medical University of Vienna, European Union, Vienna, Austria.*
E-mail: dietmar.pils@univie.ac.at

**Yunfu Cui**, *Department of Hepatopancreatobiliary Surgery, Second Affiliated Hospital of Harbin Medical University, Harbin, China.*
E-mail: yfciui777@hotmail.com

**Martin Filipits**, *Institute of Cancer Research, Department of Medicine I, Medical University of Vienna, Vienna, Austria.*
E-mail: martin.filipits@meduniwien.ac.at

Thank you very much for your attention and consideration. We are looking forward to your response.

Kind regards.

Sincerely

Cui Yukun

Email: ycui27@gmail.com